

Citation: Prendergast CT, Sanin DE, Mountford AP (2016) Alternatively Activated Mononuclear Phagocytes from the Skin Site of Infection and the Impact of IL-4R α Signalling on CD4⁺T Cell Survival in Draining Lymph Nodes after Repeated Exposure to *Schistosoma mansoni* Cercariae. PLoS Negl Trop Dis 10(8): e0004911. doi:10.1371/journal. pntd.0004911

Editor: Stephen John Davies, Uniformed Services University, UNITED STATES

Received: May 31, 2016

Accepted: July 18, 2016

Published: August 9, 2016

Copyright: © 2016 Prendergast et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and the MicroArray raw data files, as well as the individual process files, are available from GEO (Gene expression omnibus) accession number GSE81938: http://www.ncbi.nlm.nih.gov/geo/query/ acc.cgi?token=szqricacpfczvcb&acc=GSE81938.

Funding: This project was funded by a Project Grant from the Wellcome Trust awarded to APM (Grant number: 092745/Z/10/Z). CTP was funded by the Wellcome Trust. The funders had no role in study **RESEARCH ARTICLE**

Alternatively Activated Mononuclear Phagocytes from the Skin Site of Infection and the Impact of IL-4Ra Signalling on CD4⁺T Cell Survival in Draining Lymph Nodes after Repeated Exposure to *Schistosoma mansoni* Cercariae

Catriona T. Prendergast^{**a}, David E. Sanin^{**b}, Adrian P. Mountford*

Centre for Immunology and Infection, Department of Biology, University of York, York, United Kingdom

¤a Current address: Institute of Infection, Immunity and Inflammation, University of Glasgow, Glasgow, United Kingdom

¤b Current address: Department of Immunometabolism, Max Planck Institute of Immunobiology and Epigenetics, Freiburg, Germany

* adrian.mountford@york.ac.uk

Abstract

In a murine model of repeated exposure of the skin to infective Schistosoma mansoni cercariae, events leading to the priming of CD4 cells in the skin draining lymph nodes were examined. The dermal exudate cell (DEC) population recovered from repeatedly (4x) exposed skin contained an influx of mononuclear phagocytes comprising three distinct populations according to their differential expression of F4/80 and MHC-II. As determined by gene expression analysis, all three DEC populations (F4/80⁻MHC-II^{high}, F4/80⁺MHC-II^{high}, F4/80⁺MHC-II^{int}) exhibited major up-regulation of genes associated with alternative activation. The gene encoding RELMα (hallmark of alternatively activated cells) was highly upregulated in all three DEC populations. However, in 4x infected mice deficient in RELMa, there was no change in the extent of inflammation at the skin infection site compared to 4x infected wild-type cohorts, nor was there a difference in the abundance of different mononuclear phagocyte DEC populations. The absence of RELMa resulted in greater numbers of CD4⁺ cells in the skin draining lymph nodes (sdLN) of 4x infected mice, although they remained hypo-responsive. Using mice deficient for IL-4Ra, in which alternative activation is compromised, we show that after repeated schistosome infection, levels of regulatory IL-10 in the skin were reduced, accompanied by increased numbers of MHC-II^{high} cells and CD4⁺ T cells in the skin. There were also increased numbers of CD4⁺ T cells in the sdLN in the absence of IL-4Ra compared to cells from singly infected mice. Although their ability to proliferate was still compromised, increased cellularity of sdLN from 4x IL-4RaKO mice correlated with reduced expression of Fas/FasL, resulting in decreased apoptosis and cell death but increased numbers of viable CD4⁺ T cells. This study highlights a mechanism

design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

through which IL-4R α may regulate the immune system through the induction of IL-10 and regulation of Fas/FasL mediated cell death.

Author Summary

In areas endemic for schistosomiasis, repeated exposure to infective cercariae is a frequent occurrence, and repeated exposure of murine skin to Schistosoma mansoni resulted in CD4⁺ T cells becoming hypo-responsive. Here potential contributory mechanisms were investigated. In the skin infection site, three mononuclear phagocyte populations were identified (tissue macrophages, dendritic cells, and macrophages) which exhibited up-regulation of genes associated with alternative activation, in particular the gene encoding RELMa. However, in repeatedly infected mice deficient in RELMa, there was no change in the abundance of mononuclear phagocytes in the skin, and CD4⁺ cells in the skin draining lymph nodes remained hypo-responsive. In mice deficient for IL-4R α , required for alternative activation, levels of dermal regulatory IL-10 were reduced and there was an increase in the abundance of antigen presenting MHC-II^{high} cells, which was accompanied by increased numbers of CD4⁺ T cells. Although the absence of IL-4Ra did not translate into increased CD4⁺ cell responsiveness, they exhibited lower expression of Fas/FasL, resulting in decreased apoptosis/cell death and increased cell viability. This study highlights a mechanism through which IL-4R α may regulate the immune system through the induction of IL-10 and regulation of Fas/FasL mediated cell death.

Introduction

Schistosomiasis is a debilitating disease that develops following percutaneous infection with the parasitic helminth *Schistosoma spp* $[\underline{1}, \underline{2}]$. The disease affects approximately 230 million people worldwide with infection occurring when the skin is exposed to the free-swimming tissue invasive cercariae $[\underline{3}]$. Since the infective stage of the parasite is often present in water used for domestic purposes, individuals living in areas endemic to schistosomiasis are at risk of repeated infections.

In order to investigate the effect of repeated infection with schistosome cercariae on the immune response, we developed an experimental model whereby mice were exposed via their pinnae once (1x), or repeatedly (4x), to doses of infective *S. mansoni* cercariae [4]. It was found that after 4x, compared to 1x, exposures there were major changes in the cell populations within the skin site of infection such that eosinophils, macrophages, dendritic cells (DCs), neutrophils, mast cells, CD4⁺ T cells and keratinocytes were all increased after 4x infections [4–7]. Moreover, repeated infections resulted in the development of CD4⁺ T cell hypo-responsiveness in the skin-draining lymph nodes (sdLN), as well as decreased immunopathology in the liver generated in response to eggs released by adult worms [4]. CD4⁺ T cells in the sdLN from 4x mice had reduced ability to proliferate and secrete cytokines in response to larval schistosome antigens, and this was shown to be IL-10 dependent [6, 8].

Exposure of the skin to repeated doses of schistosome cercariae caused major changes in the local cytokine environment, particularly the levels of IL-4, IL-13 and IL-10, and it was proposed that immune responses in the skin involved mononuclear phagocytes that were alternatively activated [4]. The term alternative activation conventionally describes macrophages under the influence of IL-4 and IL-13 [9–12]. Alternatively activated macrophages have been

given the term M2, or subgroupings thereof M2a-c [13], whilst the term alternative activation has also been used in the context of DCs [14]. Parasitic helminth infections are often associated with the development of alternatively activated cells and polarized Th2-type immune responses [15]. Together, these cell populations are linked to the development of wound healing, which accompanies the response to tissue invasive helminths [16, 17]. Therefore, it is likely that there will be substantial wound healing following the repeated percutaneous invasion of the skin by schistosome cercariae.

In the current study, we set out to investigate mononuclear phagocyte cell populations in the skin infection site following repeated exposure to schistosome larvae, and found evidence for a gene expression profile associated with alternative activation. Subsequently, we focused our investigation upon the impact of IL-4R α which is a receptor for both IL-4 and IL-13 essential for the alternative activation of macrophages [9, 10], and Resistin-like molecule α (RELM α) which is a key marker of alternative activation and is abundant during Th2 immune responses in allergic lung inflammation and helminth infection [18–21]. Consequently, we hypothesized that IL-4R α and RELM α might play an important role in the response following repeated exposures to infectious *S. mansoni* cercariae.

Materials and Methods

Ethics statement

Experiments were carried out in accordance with the United Kingdom Animals Scientific Procedures Act 1986 and with the approval of the University of York Ethics Committee (PPL 60/4340).

Mice

C57BL/6 wild-type (WT), IL-4R α deficient (Il-4r $\alpha^{-/-}$; IL-4R α KO) [22], and Resistin-like molecule α deficient (*Retln\alpha^{-/-}*; Relm α KO) [23], mice were bred and housed at the University of York. Both IL-4R α KO and Relm α KO mice on C57BL/6 background were kind gifts from Professor Judith Allen, University of Edinburgh. Age and sex-matched mice (between 6 and 10 weeks) were used for all experiments.

S. mansoni parasites and skin infection

The life cycle of a Puerto Rican strain of *S. mansoni* was maintained at the University of York. Both pinnae of mice were percutaneously exposed to 150 *S. mansoni* cercariae either once only (1x), or four times in total (4x) on a once-a-week basis between day 0 to day 21, as described previously [4, 24]. Pinnae and skin-draining lymph nodes (sdLN; auricular lymph nodes) were harvested 4 days after the final infection. Skin infection via the pinnae results in a 50% penetration rate [24], therefore a dose of approximately 75 cercariae per pinna is achieved. Pinnae inflammation was measured using a dial gauge micrometer (Mitutoyo, Japan).

Isolation of dermal exudate cells (DEC)

Pinnae were collected and briefly exposed to 70% ethanol to sterilize before being air-dried. They were then split along the central cartilage into two halves, and floated on the surface of complete RPMI media (RPMI-1640 (Gibco, Paisley, UK) containing 10% heat inactivated FCS (Biosera, Uckfield, UK), 2 mM L-Glutamine, 1% Pen/Strep (both Gibco) and 50 μ M 2-mercaptoethanol (Sigma-Aldrich, Gillingham, UK) overnight at 37°C 5% CO₂, to obtain the dermal exudate cells (DEC) as previously described [4, 24, 25]. After overnight *in vitro* culture, the tissue samples were discarded, while the culture supernatants were spun at 1000 xg for 7 minutes

at 4°C to recover the DEC. The skin biopsy culture supernatants were frozen at -20°C prior to subsequent analysis by ELISA. DEC were re-suspended in complete RPMI, numbers determined by trypan blue exclusion, and then subjected to either analysis by flow cytometry, or separated into groups by fluorescence-activated cell sorting (FACS).

Cytokine analysis by ELISA

The amounts of released IL-4, IL-10, IL-12p40 present in the skin biopsy culture supernatants were determined using DuoSet enzyme-linked immunosorbent assay (ELISA) kits (R&D Systems, Abingdon, UK).

In vivo lymph node cell proliferation

Mice received 1 mg bromodeoxyuridine (BrdU; Sigma-Aldrich) via daily intraperitoneal (i.p.) injection for the final 4 days prior to harvest of the sdLN in order to determine *in vivo* CD4⁺ cell proliferation [8]. Cells recovered from the sdLN were initially blocked with anti-CD16/32 monoclonal antibodies (mAbs; eBioscience, Hatfield, UK,) in goat serum (Sigma-Aldrich), and later labelled using anti-CD3 and anti-CD4 mAbs (both eBioscience) in PBS supplemented with 1% FCS (FACS Buffer). Cells were then washed in FACS Buffer, incubated in 1x Fixation/ Permeabilization buffer (eBioscience) for one hour at 4°C, washed again, and then incubated at 37°C in 100 µg DNase (Sigma-Aldrich) for 1 hour. After a final wash in FACS buffer, cells were labelled for 45 minutes at room temperature with anti-BrdU mAb, or rat IgG1 isotype control mAb (both eBioscience), in 1x permeabilization buffer, according to the manufacturer's instructions.

Flow cytometry

DEC were first incubated using Fixable Live/Dead Aqua stain (Life Technologies, Paisley, UK), blocked with anti-CD16/32 mAbs (eBioscience) in goat serum (Sigma Aldrich), and subsequently labeled with the following mAbs conjugated to fluorescent labels: anti-CD45, anti-F4/80, anti-MHC-II (IA-IE), anti-Fas, anti-FasL, anti-CD3 and anti-CD4 (all eBioscience). Flow cytometry data was acquired on the Cyan ADP, or the BD LSR Fortessa analyzer (Beckman Coulter, London, UK). Data was analyzed using FlowJo Software v7.6.5 (Tree Star Inc, Oregon Bio, Oregon US).

Annexin V assay

After surface staining for anti-CD3 and anti-CD4, sdLN cells were washed in cold PBS supplemented with 1x annexin V binding buffer (eBioscience), and incubated for 15 minutes with anti-annexin V FITC at room temperature. Cells were then washed in annexin V binding buffer and resuspended for analysis in annexin V binding buffer. Propidium iodide (PI) (eBioscience) was added directly before acquiring the data.

FACS-sorting of DEC and microarray analysis

DEC obtained from infected pinnae in three independent experiments (12–18 mice each) recovered on day 4 after infection with *S. mansoni* cercariae were pooled and labelled with anti-MHC-II (IA-IE) (clone # M5/114) and anti-F4/80 (clone # BM8) mAbs. Cell populations which were F4/80⁺MHC-II^{high}, F4/80⁺MHC-II^{lo}, or F4/80⁻MHC-II^{high} were recovered by FACS (MoFlo Astrios, Beckman Coulter) and RNA extracted from sorted DEC populations using TRIzol (Life Technologies). RNA was quantified using a Nanodrop (Thermo Scientific, Waltham, USA) and quality checked using a Bioanalyzer (Agilent Technologies, Santa Clara,

USA). Staff at the Technology Facility at the University of York, York, UK, prepared and carried out microarray analysis of purified RNA, including sample labelling and hybridisation, using the Agilent SurePrint system (Agilent Technologies).

Bioinformatics

GeneSpring (Agilent Technologies) was used to normalize microarray data, calculate fold differences, prepare dendrograms of cell populations and establish significant (p < 0.05) differentially expressed genes.

Statistics

Statistical analyses were performed using a one-way analysis of variance (ANOVA) and Tukey's multiple comparisons test using GraphPad Prism v6 software (GraphPad Software Inc, San Diego, California. USA).

Results

Genes associated with alternative activation were highly up-regulated in DEC recovered from the skin after repeated exposures to *S. mansoni* cercariae

Dermal exudate cells (DEC) were recovered from *in vitro* cultured skin biopsies after exposure to either a 1x dose, or 4x doses, of *S. mansoni* cercariae in order to characterize genes that were differentially regulated four days after the final infection. Cells which were F4/80⁺MHC-II⁻, shown to be SiglecF⁺ eosinophils [4, 12] were excluded from the analysis as we previously showed that the absence of these cells had no effect on the development of CD4⁺ cell hyporesponsiveness in the sdLN [7]. Therefore, our microarray analysis focused upon cells of the mononuclear phagocyte system (i.e. DCs and macrophages). These cells distributed into three discrete populations: F4/80⁻MHC-II^{high} cells (denoted 'R4') were classed as DCs, F4/80⁺MHC-II^{int} cells (denoted 'R4A') were termed tissue resident macrophages, whilst F4/80⁺MHC-II^{int} cells (denoted 'R3') were classified as macrophages [6] as shown in Fig 1A.

Clustering analysis of microarray data obtained from biological replicates of the three sorted DEC populations is shown as dendrograms with corresponding selected heat maps for cells from 1x and 4x infected groups of mice (Fig 1B and 1C). The heat maps displayed highlight a section from the start of the entire microarray heat map, whilst the dendrograms are based upon analysis of all identified genes to yield an overall comparison. This revealed clear clustering patterns within each of the defined cell populations (i.e. R4, R4A, and R3) validating our gating strategy. Based on the dendrogram analysis of the microarray data from the 1x and 4x DEC populations, gene expression profiles within the R4A tissue resident macrophages were more closely associated to R3 macrophages than to R4 DCs (Fig 1B and 1C). Transcriptional distinctions between the R4A and R3 populations were reduced after 4x infection (Fig 1C).

Many identified genes found in the three sorted DEC populations were differentially up-regulated in the 4x compared to the 1x samples, and were linked to alternative activation (e.g. *retnla*, *chi3l3*, *chi3l4*, *ccl24*, *cd209* and *cd163* [11, 26–28]) (Fig 1D). *Retnla* (encoding RELMα) was one of the most highly up-regulated genes in 4x DCs (R4; x121-fold), 4x tissue macrophages (R4A; x16-fold) and 4x macrophages (R3; x80-fold), compared to their 1x counterpart cell populations. Similarly, *chi3l3* was up-regulated in both 4x tissue macrophages (R4A; x31-fold) and 4x macrophages (R3; x31-fold), whilst *ccl24* was up-regulated in all three cell populations 24-33-fold. The gene for IL-4 was up-regulated in 4x macrophages (R3; x22-fold), whilst *il4ra* was also up-regulated in these cells (R3; x6-fold). Other genes which featured in the



Calca (11) Pax2 (11) Ear12 (5) Cdh1 (5) Fn1 (14) Fig 1. Genes related to alternative activation cell phenotype are up-regulated in dermal exudate cells (DEC) populations after repeated infection. A. Representative flow cytometry dot plots for DEC showing the distribution of cells based upon their expression of F4/80 and MHC-II. Cells were gated into R4, R4A and R3 populations recovered from 1x (left) and 4x (right) infected mice. Values in bold are the percent positive cells found within each gate expressed as a proportion of the total DEC recovered. Cells were sorted by

Prg2 (17)

Cd209e (16)

Hbegf (15)

Cd81 (14)

Etv1 (5)

Igf1 (5)

Rcn3 (5)

Prg2 (16)

Cbr2 (15)

Ear2 (14)

fluorescent activated cell sorting (FACS), and RNA from the sorted R4, R4A and R3 cell populations was applied to microarray analysis. Heat maps showing the clustering of genes within each sorted DEC population for three biological replicates recovered after **B**. a single (1x) infection, and **C**. repeated (4x) infections. **D**. Identity of the top 15 up-regulated genes in each population after 4x compared to 1x infection. Number in brackets represents the fold up-regulation of that gene in the sorted cell population from 4x infected mice. Grey shaded genes represent those that are associated with alternative activation.

doi:10.1371/journal.pntd.0004911.g001

top 20 up-regulated genes included those associated with tissue destruction/wound healing such as *igf1* (x8-30-fold) and *fn1* (x14-23-fold).

Inflammation of the skin infection site in the absence of IL-4R α and RELM α following 4x schistosome infection

Given the significant changes in gene expression associated with IL-4R α and RELM α , their roles in the early stage immune response were investigated using *Il-4r\alpha^{-/-}*, or *Retln\alpha^{-/-}* mice. After 4x exposures to infective *S. mansoni* cercariae, pinnae thickness of WT mice, as well as those deficient in IL-4R α , was significantly increased compared to 1x exposure (Fig 2A; p<0.0001 and p<0.05 respectively). However, the thickness of the pinnae infection site was comparable between WT and IL-4R α KO mice, irrespective of the infection regime (Fig 2A; p>0.05). Similarly, whilst Relm α KO mice exposed to 4x doses of cercariae displayed significantly enhanced levels of skin inflammation compared to naive and 1x Relm α KO animals (all p<0.0001), the levels of inflammation were comparable with those of their 4x WT cohorts (Fig 2B; all p>0.05).

Increased levels of IL-4, IL-12p40 and IL-10 released by skin biopsies cultured *in vitro* in the absence of added antigen were detected in the pinnae of 4x compared to 1x WT mice (Fig 2C-2H; p<0.05–0.0001). In the absence of IL-4R α , the levels of IL-4 in 4x IL-4R α KO mice were comparable to those in 4x WT samples (Fig 2C), on the other hand, the levels of pro-inflammatory IL-12p40 in 4x IL-4R α KO mice were significantly increased compared to 4x WT samples (Fig 2D; p<0.001). The levels of regulatory IL-10 were low in 4x IL-4R α KO mice (Fig 2E), resulting in a significant reduction in the quantities of IL-10 being released compared to 4x WT mice (Fig 2E; p<0.0001). The absence of RELM α had no effect on the production of IL-4, or IL-12p40 in 1x and 4x mice compared to their WT cohorts (Fig 2F and 2G; p>0.05), however similar to 4x IL-4R α KO mice, IL-10 production was significantly lower in 4x Relm α KO mice produce similar amounts of IL-4 after 1x and 4x infections, IL-4R α KO mice have a more pro-inflammatory phenotype (i.e. increased IL-12p40) and both KO strains had decreased IL-10 production compared to WT mice.

The absence of IL-4R α , but not RELM α , increases the abundance of mononuclear phagocytes recovered from the skin

As expected, significantly greater numbers of DEC were recovered from 4x WT compared to 1x WT mice (Fig 3A and 3B; p<0.01), although as with pinnae thickness, DEC numbers in 1x and 4x infected IL-4R α KO mice were similar compared to their WT cohorts (Fig 3A; p>0.05). The number of DEC recovered from pinnae biopsies of 4x Relm α KO compared to 4x WT mice was also not significantly different (Fig 3B; p>0.05).

The abundance of F4/80⁺ antigen presenting cells may be critical to the development of CD4⁺ T cell hypo-responsiveness, particularly as it has been shown that F4/80⁺ alternatively activated macrophages depress CD4⁺ cell responses following infection with filarial parasites [29]. Here, we show there were no significant changes in the number of F4/80⁻MHC-II^{high}

PLOS | NEGLECTED TROPICAL DISEASES



Fig 2. Inflammation of the skin infection site in the absence of IL-4R α and RELM α following 4x schistosome infection. A. Pinnae thickness in naïve, 1x and 4x infected WT and IL-4R α KO mice and, **B**. in naïve, 1x and 4x infected WT and Relm α KO mice on day 4 after the final infection. Symbols are values for individual mice; horizontal bars are means ±SEM (n = 4–5 mice). **C-H.** Production of IL-4, IL-12p40 and IL-10 by skin biopsies (cultured in the absence of exogenous parasite antigen) from groups of WT, IL-4R α KO and Relm α KO mice. Cytokine production in the overnight culture supernatants was determined by ELISA (pg/ml); n = 4–5 mice per group. * = p<0.05; ** = p<0.01; *** = p<0.001, **** = p<0.0001, n.s. = p>0.05 as determined by ANOVA and Tukey's post test analysis.

doi:10.1371/journal.pntd.0004911.g002

(R4), F4/80⁺MHC-II^{high} (R4A), or F4/80⁺MHC-II^{int} (R3) cells in 4x WT compared to 1x WT mice (Fig 3C–3E; all p>0.05). However, in the absence of IL-4R α signaling, there were large increases in the numbers of these three cell types in 4x IL-4R α KO compared to 1x IL-4R α KO mice (Fig 3C–3E; p<0.05–0.01). Moreover, the number of R4 and R4A cells, which express high levels of MHC-II, were significantly greater in 4x IL-4R α KO than in 4x WT cohorts (Fig 3C and 3D; p<0.05–0.01), although the change in the number of R3 macrophages which

PLOS | NEGLECTED TROPICAL DISEASES



Fig 3. The absence of **IL-4Rα** and **Relmα** affects the phenotype of mononuclear phagocytes recovered from the skin. Total number of viable DEC recovered from pinnae of WT and **A**. **IL-4RαKO**, and **B**. RelmαKO mice. Absolute numbers of **C & F**. F4/80⁻MHC-II^{high} DCs (R4), **D & G**. F4/80⁺MHC-II^{high} tissue resident macrophages (R4A), and **E & H**. F4/80⁺MHC-II^{int} macrophages (R3) in 1x and 4x infected WT, IL-4RαKO and RelmαKO mice. **I**. Absolute numbers of CD3⁺CD4⁺ T cells in the DEC of 1x and 4x infected WT and IL-4RαKO mice. Symbols are values for individual mice; horizontal bars are means ±SEM (n = 4–5 mice per group). n.s. denotes 'not significant' p>0.05; * = p<0.05; ** = p<0.01; *** = p<0.001; **** = p<0.0001 as determined by ANOVA and Tukey's post-test analysis, or Student's t test.

doi:10.1371/journal.pntd.0004911.g003

express intermediate levels of MHC-II in 4x IL-4R α KO and 4x WT mice was not statistically different (Fig 3E; p>0.05). In Relm α KO mice, the numbers of R3, R4 and R4A DEC were similar to those in WT cohorts, and there was no significant difference in the number of these two cell types after 4x compared to 1x infection (Fig 3F-3H; p>0.05).

Increased numbers of MHC-II^{high} R4 and R4A cells in 4x infected IL-4R α KO mice could support a stronger T cell response, both in the skin site of infection and down-stream in the draining lymphoid tissue [30, 31]. This might be particularly relevant when coupled with low levels of IL-10 production in the skin, a cytokine that we recently showed to be fundamental in regulating CD4⁺ T cell numbers [6, 8]. Indeed, in the absence of IL-4R α signaling, significantly greater numbers of CD3⁺CD4⁺ T cells were present in the skin infection site of 4x IL-4R α KO, compared to 4x WT mice (Fig 3I; p<0.0001). Therefore, we subsequently investigated the changes in the numbers, viability, and/or responsiveness of CD4⁺ cells in the sdLN.

Increased CD4⁺ cellularity in the sdLN of 4x infected mice in the absence of IL-4R α and RELM α but no reversal of hypo-responsiveness

After 4x infections, the sdLN of WT mice had significantly fewer cells compared to 1x WT mice (Fig 4A; p<0.05). However, in the absence of IL-4R α , cellularity was increased, resulting in comparable cell numbers between 1x and 4x infected IL-4R α KO mice which were not significantly different (Fig 4A; p>0.05). Similarly, total sdLN cellularity was also increased in 4x Relm α KO compared to 4x WT mice (Fig 4B; p<0.01), resulting in the number of sdLN cells from 4x and 1x Relm α KO mice being not significantly different (Fig 4B; p>0.05).

There were also decreased numbers of $CD4^+$ T cells in the sdLN of 4x compared to 1x WT mice (Fig 4C; p<0.05), although the number of $CD4^+$ T cells in 4x compared to 1x IL-4R α KO mice was not significantly reduced (Fig 4C; p>0.05). Therefore, in comparison there were significantly more $CD4^+$ T cells in 4x IL-4R α KO than in their 4x WT counterparts (Fig 4C; p<0.05). The numbers of $CD4^+$ T cells in the sdLN of 1x and 4x Relm α KO mice was also similar (Fig 4D; p>0.05) leading to significantly greater numbers of $CD4^+$ T cells being detected in the sdLN of 4x Relm α KO compared to 4x WT mice (Fig 4D; p<0.05). This showed that the absence of either IL-4R α , or RELM α , leads to increased number of $CD4^+$ T cells in the sdLN following repeated infection comparable to the levels seen in 1x mice.

It was proposed that the increased number of CD4⁺ T cells in sdLN of 4x IL-4R α and 4x RelmaKO mice could be due to a reversal in the development of hypo-responsiveness normally seen in 4x WT mice. In fact, the total number of BrdU⁺CD4⁺ cells was slightly greater in 4x IL- $4R\alpha KO$ mice compared to the 4x WT cohort group (Fig 4E; p<0.05), although the number in 4x Relm α KO versus 4x WT mice was not significantly different (Fig 4F; p>0.05). While ~15– 20% of the CD4⁺ T cells in the sdLN from 1x WT mice were BrdU⁺ and had therefore proliferated *in vivo*, only \sim 7% from 4x WT mice were BrdU⁺ (Fig 4G; p<0.0001) confirming the establishment of CD4⁺ T cell hypo-responsiveness. However, although there were a slightly greater number of BrdU⁺ cells in 4x IL-4R α KO mice (Fig 4E), the proportion of BrdU⁺ cells in 4x IL- $4R\alpha KO$ mice remained significantly lower than in 1x IL- $4R\alpha KO$ mice (Fig 4G; p<0.05) demonstrating that the absence of IL-4R α does not restore proliferation of CD4⁺ T cells in the sdLN of 4x mice to the levels seen in 1x WT mice. A similar situation was observed in 4x Relm α KO mice, where the proportions of BrdU⁺ cells in 4x Relm α KO compared to 4x WT mice were not significantly different (Fig 4H; p>0.05) and both exhibited CD4 T cell hyporesponsiveness *in vivo* compared to their 1x cohorts (Fig 4H; p<0.0001). This shows that the absence of RELM α also does not restore CD4⁺ T cell proliferation in the sdLN after repeated exposures of the skin to S. mansoni cercariae.



Fig 4. Increased CD4⁺ cellularity of sdLN from 4x infected mice in the absence of IL-4Rα and RELMα but not a reversal of CD4⁺ cell hypo-responsiveness. Number of A & B. total cells, and C & D. CD4⁺ T cells in the sdLN of naïve, 1x and 4x infected WT, IL-4RαKO, and RelmαKO mice. E & F Number of CD4⁺ BrdU⁺ cells, and G & H. percentage of CD4⁺ T cells that are BrdU⁺ in the sdLN of naïve, 1x and 4x infected WT mice compared to IL-4RαKO and RelmαKO mice. Symbols are values for individual mice; horizontal bars

are means \pm SEM (n = 4–5 mice per group). n.s. denotes 'not significant' p>0.05; * = p<0.05; ** = p<0.01; *** = p<0.001; **** = p<0.001 as determined by ANOVA and Tukey's post-test analysis, or Student's t test.

doi:10.1371/journal.pntd.0004911.g004

Expression of Fas/FasL and viability of CD4⁺ cells in the sdLN depends upon the presence of IL-4R α

Previously, we reported that CD4⁺ T cell hypo-responsiveness in the sdLN observed after repeated infection is dependent on IL-10 [8]. This was due to increased CD4⁺ T cell activation accompanied by decreased death and apoptosis of the CD4⁺ T cell population in the sdLN. In the current study, we show that both IL-4R α KO and Relm α KO mice had decreased IL-10 production in the skin and increased cellularity in the sdLN after 4x infection. As RELMa expression is dependent on IL-4 signaling [32], we restricted further analysis of the CD4⁺ cell population to cells in the sdLN of 1x versus 4x IL-4RaKO mice. We observed that surface protein expression of both Fas and FasL increased in CD4⁺ T cells recovered from 4x WT compared to 1x WT mice but was significantly decreased in 4x IL-4RaKO compared to 4x WT mice (Fig 5A and 5B; p<0.05 and p<0.0001). In addition, whereas in WT mice, there was a significant decrease in the proportions of AnnV⁻PI⁻ viable CD4⁺ T cells in 4x compared to 1x mice (Fig 5C; p < 0.01), the absence of IL-4R α signaling resulted in significantly greater proportions of AnnV⁻PI⁻ viable CD4⁺ T cells in 4x IL-4R α KO compared to 4x WT mice (Fig 5C; p<0.01). This caused there to be no significant difference between the viability of CD4⁺ T cells from 4x IL-4R α KO mice compared to either 1x WT, or 1x IL-4R α KO mice (Fig 5C; p>0.05). The increase in the proportion of AnnV⁻PI⁻ viable CD4⁺ T cells in 4x IL-4R α KO mice was accompanied by the detection of significantly fewer AnnV⁺PI⁻ apoptotic CD4⁺ T cells (Fig 5D; p < 0.05), as well as fewer AnnV⁺PI⁺ dead CD4⁺ T cells, compared to 4x WT mice (Fig 5E; p < 0.01). Thus, it appears that a reduction in IL-4R α signaling facilitates CD4⁺ T cell survival. This increased survival could explain why the number of CD4⁺ T cells in the sdLN of 4x IL-4RαKO mice is not significantly reduced after repeated exposure to schistosome cercariae which contrasts with the situation in 4x WT mice where there is a significant reduction in number of CD4⁺ T cells when compared to their 1x WT counterparts.

Discussion

Here we demonstrate that three discrete mononuclear phagocyte cell populations were present in DEC recovered from the skin infection site of mice exposed to repeated doses of schistosome cercariae. Moreover, all three populations exhibited an alternatively activated phenotype as judged by microarray analysis of DEC. For example, *Retnla* (encoding RELM α) which is a key marker of alternatively activated macrophages [33, 34] was one of the most highly up-regulated genes in 4x compared to 1x DCs, tissue macrophages, and macrophages. Several other genes also linked to alternative activation were up-regulated in the three discrete 4x DEC populations including chi3l3 and chi3l4 (encoding chitinase-like molecules YM-1 and YM-2 [12, 33]). The gene for IL-4 was also up-regulated in the macrophage DEC population after repeated infections, and IL-4 has been shown to be produced by alternatively activated/type-II macrophages using both human and mouse cells [35, 36]. Other up-regulated genes linked to alternative activation were mrc1 and cd209 which encode the C-type lectins for the mannose receptor CD206 and DC-SIGN CD209 [9], whilst expression of *cd163* encoding the scavenger receptor CD163 when used in combination with *chi3l3* and *Retnla* [11, 12, 37] also supports the identification of the three discrete 4x DEC populations as being alternatively activated. Conversely, Arg1 which we had previously reported to be up-regulated in DEC from 4x mice [4], was not >2 fold up/down-regulated in the any of the 3 sorted cell populations, whilst *Clec10a* of the C-type

PLOS | NEGLECTED TROPICAL DISEASES



Fig 5. Expression of Fas/FasL and viability of CD4⁺ T cells in the sdLN is enhanced in the absence of IL-4R α . Percentage of CD4⁺ T cells that are **A.** Fas⁺ and **B.** FasL⁺ in 1x and 4x infected WT compared to IL-4R α KO mice. **C-E** Percentage of viable (AnnV⁻PI⁻), apoptotic (AnnV⁺PI⁻) and dead (AnnV⁺PI⁺) CD4⁺ T cells in the sdLN of 1x and 4x infected mice as determined by an AnnexinV assay. n = 3–4 mice per group; n.s. denotes 'not significant' p>0.05; * = p<0.05; ** = p<0.01; **** = p<0.001 as determined by ANOVA and Tukey's post-test analysis.

doi:10.1371/journal.pntd.0004911.g005

lectin/C-type lectin-like domain (CTL/CTLD) superfamily, associated with alternative activation [9] was only 2.2 fold up-regulated in the DC population. On the other hand, wound healing associated genes such as *igf-1* encoding Insulin-like growth factor 1 suggested to be involved in resolving tissue damage following helminth infection [38], and *fn1* encoding fibronectin produced during the resolution of tissue damage [39] were both highly up-regulated. This provides evidence supporting an evolutionary link between alternative activation in mononuclear phagocytes and the resolution of tissue damage caused by helminth infection [16, 17, 38]. Indeed, we have shown that schistosome cercariae cause significant tissue disruption as they invade through the skin [3], and therefore tissue repair of the skin following repeated exposure to invasive schistosome cercariae should be expected.

The discrete DEC mononuclear phagocyte populations all express genes typically associated with alternative activation [9–12]. This provides a more in-depth analysis of DEC from 4x mice in which mononuclear phagocytes appeared to switch from classically-activated to alternatively-activated commensurate with up-regulated mRNA transcripts for Ym1 and RELM α but only low levels of iNOS and IFN γ [4]. Nevertheless, whilst the R4, R4A, and R3 mononuclear phagocytes all appear to be alternatively activated, subtle qualitative differences in the expression of specific genes between the three DEC populations underline the likely heterogeneity/plasticity of these mononuclear phagocytes which may have different and/or overlapping functional roles *in vivo* [40–42].

In the context of repeat infection of the skin with S. mansoni cercariae, DEC recovered from the site of infection had a pronounced increase in the expression of RELM α but its absence had no effect on the extent of skin inflammation (i.e. pinnae thickness), or on the numbers of mononuclear phagocyte DEC populations recovered from the skin infection site (Fig 3F-3H). The absence of RELM α did however result in a slight reduction in IL-10 production in the skin alongside an increase in the numbers of CD4⁺ cells in the sdLN such that they were as abundant as in 1x RELMaKO mice and were more numerous than in 4x WT cohorts. However, the absence of RELM α did not affect the development of CD4⁺ T cell hypo-responsiveness in the sdLN of 4x mice. In contrast, a separate study showed that RELMa KO mice infected with Nippostrongylus brasiliensis exhibit enhanced intestinal and pulmonary pathology accompanied by expulsion of the parasite indicating a regulatory role for RELMa by dampening normally protective strong Th2-dependent responses [32, 43]. Moreover, pulmonary immune granulomas to injected S. mansoni eggs were enhanced in the absence of RELM α [44], as were hepatic immune granulomas formed at the chronic phase of schistosome infection [32]. Nonetheless, these previous studies were performed when the immune response was skewed towards a Th2 phenotype [21]. In contrast, the response investigated in our study occurs at an early phase of infection, prior to egg-induced Th2 biased immunopathology, and is accompanied by cytokines with a mixed Th1 and Th2 type response [4]. Therefore, RELMa may only have a regulatory role during strong Th2 responses during the chronic phase of infection and we suggest that it has less of a role when the immune response has a mixed Th1/Th2 phenotype.

Signaling through IL-4Ra is well defined as being important in the development of alternative activation [10, 12] and in the maintenance of Th2 responses [45]. Whilst the absence of IL-4R α did not have an impact on inflammation at the skin site of infection, nor on the number of DEC, its absence significantly increased the numbers of cells expressing high levels of MHC-II. In addition, increased release of pro-inflammatory IL-12p40 in the absence of IL-4R α was accompanied by decreased levels of regulatory IL-10. Therefore, we considered it possible that the increased numbers of cells with antigen presenting potential in the skin might lead to enhanced CD4⁺ cell activity downstream in the sdLN. However, whilst the absence of IL-4R α resulted in increased numbers of CD4⁺ cells in the skin infection site and the sdLN, the cells in the sdLN remained hypo-responsive *in vivo* to stimulation with parasite antigen similar to 4x WT mice. The use of cell specific gene deficient mice, such as those expressing a Cre recombinase from the lysozyme M-encoding locus, which has been widely used in the context of IL- $4R\alpha$ function in macrophages and in the context of schistosome infection [22], would have been desirable to fully interrogate the role of particular genes on mononuclear phagocytes. However, such mice were not available during the current project. In addition, it has been reported that Il4r α excision in these mice is incomplete during inflammatory conditions [46], raising doubts about interpretation of data obtained using these mice and suggests that alternative cell specific gene animals should be sought.

Further analysis of CD4⁺ cells in the sdLN revealed that there was a significant elevation in the expression of Fas and FasL on the CD4⁺ T cells in the sdLN in 4x infected WT mice, whereas in the absence of IL-4R α the expression levels of these two molecules was not elevated and was not different from those in 1x IL-4R α mice. We also showed that skin biopsies from 4x IL-4R α KO mice, in contrast to 4x WT cohorts, released negligible quantities of IL-10 suggesting a possible role for IL-4R α in the promotion of IL-10 production. Indeed, IL-4R α signaling is required for the production of IL-10 derived from Th2 cells following infection with *N*. *brasiliensis*, thereby resulting in increased levels of regulation via IL-10 [47], although others show that IL-10 production following chronic schistosome infection can be IL-4R α -independent [48]. Here, in our study we found that IL-10 production in the skin was significantly reduced in the absence of IL-4R α , potentially resulting in decreased regulation via IL-10. Several studies have identified a link between IL-10 and Fas/FasL expression. For example in systemic lupus erythematosus (SLE), IL-10 is directly able to induce the expression of Fas on the surface of T cells resulting in increased levels of apoptosis [49–51]. Moreover, we recently showed that in IL-10KO mice, there was reduced Fas expression on CD4⁺ T cells in the sdLN following repeated schistosome infection which led to a reduction in CD4⁺ T cell death in the sdLN and consequently may have contributed to the alleviation of CD4⁺ T cell hypo-responsiveness in the absence of IL-10 [8]. Here, we suggest a novel mechanism for the regulation of the immune response through IL-4R α , which impacts both IL-10 production and antigen presenting cells numbers, which would subsequently regulate Fas and FasL expression on CD4⁺ T cells in the sdLN of 4x infected mice. Thus, IL-4R α signaling results in increased IL-10 production, increased levels of apoptotic and/or dead T cells in 4x mice and a dampening of the immune response. This link between IL-4R α and IL-10 and Fas/FasL-induced apoptosis could be a potential novel mechanism through which IL-4R α regulates the immune system.

Acknowledgments

The authors would like to acknowledge the technical help given by Ann Bamford to maintain the schistosome life cycle and providing infective material, Claire Bourke for technical assistance, and members of the Technology Facility in Department of Biology, University of York, for training and technical advice with flow cytometry and for performing and analysis of the microarray (Karen Hodgkinson, Karen Hogg, Graeme Park, Celina Whalley and Peter Ashton).

Author Contributions

Conceptualization: APM CTP.

Data curation: CTP DES.

Formal analysis: CTP DES.

Funding acquisition: APM.

Investigation: CTP DES.

Methodology: APM CTP.

Project administration: APM.

Supervision: APM.

Validation: APM CTP DES.

Visualization: CTP.

Writing - original draft: CTP.

Writing - review & editing: APM DES.

References

- 1. Chitsulo L, Loverde P, Engels D. Schistosomiasis. Nat Rev Microbiol. 2004; 2(1):12–3. PMID: 15035004
- Colley DG, Bustinduy AL, Secor WE, King CH. Human schistosomiasis. *Lancet*. 2014; 383 (9936):2253–64. doi: 10.1016/S0140-6736(13)61949-2 PMID: 24698483
- Paveley RA, Aynsley SA, Cook PC, Turner JD, Mountford AP. Fluorescent imaging of antigen released by a skin-invading helminth reveals differential uptake and activation profiles by antigen presenting cells. *PLoS Negl Trop Dis*. 2009; 3(10):e528. doi: <u>10.1371/journal.pntd.0000528</u> PMID: <u>19829705</u>

- Cook PC, Aynsley SA, Turner JD, Jenkins GR, Van Rooijen N, Leeto M, et al. Multiple helminth infection of the skin causes lymphocyte hypo-responsiveness mediated by Th2 conditioning of dermal myeloid cells. *PLoS Pathog*. 2011; 7(3):e1001323. doi: <u>10.1371/journal.ppat.1001323</u> PMID: <u>21445234</u>
- Bourke CD, Prendergast CT, Sanin DE, Oulton TE, Hall RJ, Mountford AP. Epidermal keratinocytes initiate wound healing and pro-inflammatory immune responses following percutaneous schistosome infection. Int J Parasitol. 2015; 45(4):215–24. doi: 10.1016/j.ijpara.2014.11.002 PMID: 25575749
- Sanin DE, Prendergast CT, Bourke CD, Mountford AP. Helminth Infection and Commensal Microbiota Drive Early IL-10 Production in the Skin by CD4+ T Cells That Are Functionally Suppressive. *PLoS Pathog.* 2015; 11(5):e1004841.
- Prendergast CT, Sanin DE, Mountford AP. CD4 T-cell hyporesponsiveness induced by schistosome larvae is not dependent upon eosinophils but may involve connective tissue mast cells. *Parasite Immunol.* 2016; 38(2):81–92. doi: 10.1111/pim.12300 PMID: 26679416
- Prendergast CT, Sanin DE, Cook PC, Mountford AP. CD4+ T cell hyporesponsiveness after repeated exposure to Schistosoma mansoni larvae is dependent upon interleukin-10. *Infect Immun.* 2015; 83 (4):1418–30. doi: 10.1128/IAI.02831-14 PMID: 25624353
- Martinez FO, Helming L, Gordon S. Alternative activation of macrophages: an immunologic functional perspective. *Annu Rev Immunol.* 2009; 27:451–83. doi: <u>10.1146/annurev.immunol.021908.132532</u> PMID: <u>19105661</u>
- Van Dyken SJ, Locksley RM. Interleukin-4- and interleukin-13-mediated alternatively activated macrophages: roles in homeostasis and disease. Annu Rev Immunol. 2013; 31:317–43. doi: <u>10.1146/</u> annurev-immunol-032712-095906 PMID: 23298208
- 11. Gordon S. Alternative activation of macrophages. *Nat Rev Immunol*. 2003; 3(1):23–35. PMID: 12511873
- Gordon S, Martinez FO. Alternative activation of macrophages: mechanism and functions. *Immunity*. 2010; 32(5):593–604. doi: <u>10.1016/j.immuni.2010.05.007</u> PMID: <u>20510870</u>
- Martinez FO, Sica A, Mantovani A, Locati M. Macrophage activation and polarization. Front Biosci. 2008; 13:453–61. PMID: <u>17981560</u>
- Cook PC, Jones LH, Jenkins SJ, Wynn TA, Allen JE, MacDonald AS. Alternatively activated dendritic cells regulate CD4+ T-cell polarization in vitro and in vivo. *Proc Natl Acad Sci U S A*. 2012; 109 (25):9977–82. doi: 10.1073/pnas.1121231109 PMID: 22660926
- Maizels RM, Pearce EJ, Artis D, Yazdanbakhsh M, Wynn TA. Regulation of pathogenesis and immunity in helminth infections. *J Exp Med*. 2009; 206(10):2059–66. doi: <u>10.1084/jem.20091903</u> PMID: 19770272
- Allen JE, Wynn TA. Evolution of Th2 immunity: a rapid repair response to tissue destructive pathogens. PLoS Pathog. 2011; 7(5):e1002003. doi: <u>10.1371/journal.ppat.1002003</u> PMID: <u>21589896</u>
- Gause WC, Wynn TA, Allen JE. Type 2 immunity and wound healing: evolutionary refinement of adaptive immunity by helminths. *Nat Rev Immunol*. 2013; 13(8):607–14. doi: <u>10.1038/nri3476</u> PMID: 23827958
- Stutz AM, Pickart LA, Trifilieff A, Baumruker T, Prieschl-Strassmayr E, Woisetschlager M. The Th2 cell cytokines IL-4 and IL-13 regulate found in inflammatory zone 1/resistin-like molecule alpha gene expression by a STAT6 and CCAAT/enhancer-binding protein-dependent mechanism. *J Immunol*. 2003; 170(4):1789–96. PMID: <u>12574343</u>
- Liu T, Jin H, Ullenbruch M, Hu B, Hashimoto N, Moore B, et al. Regulation of found in inflammatory zone 1 expression in bleomycin-induced lung fibrosis: role of IL-4/IL-13 and mediation via STAT-6. J Immunol. 2004; 173(5):3425–31. PMID: <u>15322207</u>
- Reece JJ, Siracusa MC, Scott AL. Innate immune responses to lung-stage helminth infection induce alternatively activated alveolar macrophages. *Infect Immun*. 2006; 74(9):4970–81. PMID: <u>16926388</u>
- 21. Horsnell WG, Dewals BG. RELMs in the Realm of Helminths. Trends Parasitol. 2016.
- Herbert DR, Holscher C, Mohrs M, Arendse B, Schwegmann A, Radwanska M, et al. Alternative macrophage activation is essential for survival during schistosomiasis and downmodulates T helper 1 responses and immunopathology. *Immunity*. 2004; 20(5):623–35. PMID: <u>15142530</u>
- Munitz A, Waddell A, Seidu L, Cole ET, Ahrens R, Hogan SP, et al. Resistin-like molecule alpha enhances myeloid cell activation and promotes colitis. *J Allergy Clin Immunol*. 2008; 122(6):1200–7.e1. doi: 10.1016/j.jaci.2008.10.017 PMID: 19084112
- Mountford AP, Hogg KG, Coulson PS, Brombacher F. Signaling via interleukin-4 receptor alpha chain is required for successful vaccination against schistosomiasis in BALB/c mice. *Infect Immun.* 2001; 69 (1):228–36. PMID: <u>11119510</u>

- Hogg KG, Kumkate S, Anderson S, Mountford AP. Interleukin-12 p40 secretion by cutaneous CD11c+ and F4/80+ cells is a major feature of the innate immune response in mice that develop Th1-mediated protective immunity to Schistosoma mansoni. *Infect Immun.* 2003; 71(6):3563–71. PMID: 12761141
- Loke P, Nair MG, Parkinson J, Guiliano D, Blaxter M, Allen JE. IL-4 dependent alternatively-activated macrophages have a distinctive in vivo gene expression phenotype. *BMC Immunol*. 2002; 3:7. PMID: <u>12098359</u>
- Nair MG, Gallagher IJ, Taylor MD, Loke P, Coulson PS, Wilson RA, et al. Chitinase and Fizz family members are a generalized feature of nematode infection with selective upregulation of Ym1 and Fizz1 by antigen-presenting cells. *Infect Immun*. 2005; 73(1):385–94. PMID: <u>15618176</u>
- Mantovani A, Sica A, Sozzani S, Allavena P, Vecchi A, Locati M. The chemokine system in diverse forms of macrophage activation and polarization. *Trends Immunol*. 2004; 25(12):677–86. PMID: <u>15530839</u>
- Taylor MD, Harris A, Nair MG, Maizels RM, Allen JE. F4/80+ alternatively activated macrophages control CD4+ T cell hyporesponsiveness at sites peripheral to filarial infection. *J Immunol*. 2006; 176 (11):6918–27. PMID: <u>16709852</u>
- **30.** Itano AA, Jenkins MK. Antigen presentation to naive CD4 T cells in the lymph node. *Nat Immunol.* 2003; 4(8):733–9. PMID: <u>12888794</u>
- **31.** Germain RN, Jenkins MK. In vivo antigen presentation. *Curr Opin Immunol*. 2004; 16(1):120–5. PMID: <u>14734120</u>
- Pesce JT, Ramalingam TR, Wilson MS, Mentink-Kane MM, Thompson RW, Cheever AW, et al. Retnla (relmalpha/fizz1) suppresses helminth-induced Th2-type immunity. *PLoS Pathog.* 2009; 5(4): e1000393. doi: <u>10.1371/journal.ppat.1000393</u> PMID: <u>19381262</u>
- Raes G, De Baetselier P, Noel W, Beschin A, Brombacher F, Hassanzadeh Gh G. Differential expression of FIZZ1 and Ym1 in alternatively versus classically activated macrophages. *J Leukoc Biol*. 2002; 71(4):597–602. PMID: <u>11927645</u>
- Nair MG, Cochrane DW, Allen JE. Macrophages in chronic type 2 inflammation have a novel phenotype characterized by the abundant expression of Ym1 and Fizz1 that can be partly replicated in vitro. *Immunol Lett.* 2003; 85(2):173–80. PMID: <u>12527225</u>
- La Flamme AC, Kharkrang M, Stone S, Mirmoeini S, Chuluundorj D, Kyle R. Type II-activated murine macrophages produce IL-4. *PLoS One*. 2012; 7(10):e46989. doi: <u>10.1371/journal.pone.0046989</u> PMID: <u>23071691</u>
- Pouliot P, Turmel V, Gelinas E, Laviolette M, Bissonnette EY. Interleukin-4 production by human alveolar macrophages. *Clin Exp Allergy*. 2005; 35(6):804–10. PMID: <u>15969673</u>
- Kristiansen M, Graversen JH, Jacobsen C, Sonne O, Hoffman HJ, Law SK, et al. Identification of the haemoglobin scavenger receptor. *Nature*. 2001; 409(6817):198–201. PMID: <u>11196644</u>
- Chen F, Liu Z, Wu W, Rozo C, Bowdridge S, Millman A, et al. An essential role for TH2-type responses in limiting acute tissue damage during experimental helminth infection. *Nat Med*. 2012; 18(2):260–6. doi: 10.1038/nm.2628 PMID: 22245779
- 39. Grinnell F. Fibronectin and wound healing. J Cell Biochem. 1984; 26(2):107–16. PMID: 6084665
- Hume DA, Mabbott N, Raza S, Freeman TC. Can DCs be distinguished from macrophages by molecular signatures? Nat Immunol. 2013; 14(3):187–9. doi: <u>10.1038/ni.2516</u> PMID: <u>23416664</u>
- Hume DA, Freeman TC. Transcriptomic analysis of mononuclear phagocyte differentiation and activation. *Immunol Rev.* 2014; 262(1):74–84. doi: 10.1111/imr.12211 PMID: 25319328
- Hashimoto D, Miller J, Merad M. Dendritic cell and macrophage heterogeneity in vivo. *Immunity*. 2011; 35(3):323–35. doi: <u>10.1016/j.immuni.2011.09.007</u> PMID: <u>21943488</u>
- 43. Chen G, Wang SH, Jang JC, Odegaard JI, Nair MG. Comparison of RELMalpha and RELMbeta Singleand Double-Gene-Deficient Mice Reveals that RELMalpha Expression Dictates Inflammation and Worm Expulsion in Hookworm Infection. *Infect Immun.* 2016; 84(4):1100–11. doi: <u>10.1128/IAI.01479-</u> 15 PMID: 26831469
- Nair MG, Du Y, Perrigoue JG, Zaph C, Taylor JJ, Goldschmidt M, et al. Alternatively activated macrophage-derived RELM-{alpha} is a negative regulator of type 2 inflammation in the lung. *J Exp Med*. 2009; 206(4):937–52. doi: 10.1084/jem.20082048 PMID: 19349464
- Perona-Wright G, Mohrs K, Mayer KD, Mohrs M. Differential regulation of IL-4Ralpha expression by antigen versus cytokine stimulation characterizes Th2 progression in vivo. *J Immunol*. 2010; 184 (2):615–23. doi: <u>10.4049/jimmunol.0902408</u> PMID: <u>20018622</u>
- 46. Vannella KM, Barron L, Borthwick LA, Kindrachuk KN, Narasimhan PB, Hart KM, et al. Incomplete deletion of IL-4Ralpha by LysM(Cre) reveals distinct subsets of M2 macrophages controlling inflammation and fibrosis in chronic schistosomiasis. *PLoS Pathog.* 2014; 10(9):e1004372. doi: <u>10.1371/journal.</u> <u>ppat.1004372</u> PMID: <u>25211233</u>

- **47.** Balic A, Harcus YM, Taylor MD, Brombacher F, Maizels RM. IL-4R signaling is required to induce IL-10 for the establishment of T(h)2 dominance. *Int Immunol*. 2006; 18(10):1421–31. PMID: 16940042
- Dewals BG, Marillier RG, Hoving JC, Leeto M, Schwegmann A, Brombacher F. IL-4Ralpha-independent expression of mannose receptor and Ym1 by macrophages depends on their IL-10 responsiveness. *PLoS Negl Trop Dis.* 2010; 4(5):e689. doi: <u>10.1371/journal.pntd.0000689</u> PMID: <u>20502521</u>
- Georgescu L, Vakkalanka RK, Elkon KB, Crow MK. Interleukin-10 promotes activation-induced cell death of SLE lymphocytes mediated by Fas ligand. *J Clin Invest*. 1997; 100(10):2622–33. PMID: <u>9366578</u>
- 50. Wang H, Xu J, Ji X, Yang X, Sun K, Liu X, et al. The abnormal apoptosis of T cell subsets and possible involvement of IL-10 in systemic lupus erythematosus. *Cell Immunol.* 2005; 235(2):117–21. PMID: <u>16226734</u>
- Yang X, Sun B, Wang H, Yin C, Wang X, Ji X. Increased serum IL-10 in lupus patients promotes apoptosis of T cell subsets via the caspase 8 pathway initiated by Fas signaling. *J Biomed Res.* 2015; 29 (3):232–40. doi: <u>10.7555/JBR.29.20130037</u> PMID: <u>26060447</u>